Author's response to reviews

Title: Primary pancreatic B cell lymphoma: is there a role for surgery? Case report and review of the literature

Authors:

Theodore Liakakos (theodlia@otenet.gr)
Evangelos P. Misiakos (misiakos@med.uoa.gr)
Dimitrios Tsapralis (tsapralisd@yahoo.gr)
Anastasios Macheras (anmach@med.uoa.gr)
Irene Nikolaou (lektcsurg@attikonhospital.gr)
Gabriel Karatzas (gchirclin@med.uoa.gr)

Version: 2 Date: 18 August 2007

Author's response to reviews:

Journal of Medical Case Reports
Dear Editor
It is with honor that we take the opportunity to submit our case report entitled "Primary pancreatic B cell lymphoma: is there a role for surgery? Case report and review of the literature" for consideration and possible publication in your journal.
Thank you very much
Looking forward to hearing from you.
Sincerely
Evangelos Misiakos, MD
Lecturer in Surgery
University of Athens School of Medicine